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• FAR DETECTOR:    Alec Habig
• Cosmic ray & Atmospheric ν operations continue with 99.7%

livetime this week
• NEAR DETECTOR:    Alec Habig
• Chiller maintenance completed Thur. Nov. 15
• Cosmic ray ν operations Nov. 15-Nov. 16 (no magnetic field)
• Normal Detector Operations Nov. 16-19
• Regular shift schedule begun following midnight Nov. 15
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• NEAR DETECTOR Ongoing:    Alec Habig
• high-frequency noise in Coil Current monitoring
• tracking down a few remaining single channel issues after

commissioning newly retrofitted readout electronics
• tuning hadron & muon monitors, effects of He in decay pipe not

measured yet.
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• BEAM SYSTEM EFFORTS:  Sam Childress

• Shutdown work completed Sat. afternoon, Nov. 17
• We have beam! 2+5 mixed configuration startup Sat. evening (23:45)

low intensity target scan Sat. 21:30-23:45
• Downtimes:

 3hr downtime (Sun.) due to capacitor failure in focusing quad ramps
 12-16hr downtime (Mon. 4am - evening) due to oil leak in kicker power

supply

Many thanks to all who helped make the shutdown period a success
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2+5

target
scan

• Very good intensity stability!
– Low intensity target scan at 8E11 ppp
– Operation at ~ 1.75E13 ppp
– 2.2s cycle spacing

2.2s cycle
(was 2.4s previously)
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startup beam power 
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end of 2007
shutdown
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